Author’s response to reviews

Title: Repetitive transcranial magnetic stimulation over primary motor vs non-motor cortical targets; effects on experimental hyperalgesia in healthy subjects

Authors:

Paul Sacco (psacco@liv.ac.uk)
Michael F Prior (m.f.prior@live.co.uk)
Helen Poole (MH.M.Poole@ljmu.ac.uk)
Turo J Nurmikko (tjn@liv.ac.uk)

Version: 3 Date: 1 August 2014

Author’s response to reviews:

Dear Editor,

The following revisions have been made in accordance with your instructions:

1) Please include all authors email addresses on the title page
2) In the abstract the background and objectives should be combined under the heading 'Background'
3) The Competing Interests section on page 3 should be moved so that it is just before the Authors’ Contributions
4) A conclusion section is required after the Discussion
5) The table and figure legends should be moved so that it follows the references